Author’s response to reviews

Title: The diagnostic value of homocysteine for the occurrence and acute progression of chronic obstructive pulmonary disease

Authors:

Bing Wei (fpqdfe@163.com)

Tian Tian (fsvhfi@163.com)

Yugeng Liu (dhplna@163.com)

Chunsheng Li (ikufvh@163.com)

Version: 2 Date: 10 Aug 2020

Author’s response to reviews:

Dear Editor,
Thank you for your valuable comments. We have revised the manuscript (Manuscript ID: PULM-D-20-00524R1) and sent the revised manuscript for your consideration to be published in BMC Pulmonary Medicine.
Thank you for your consideration.
We are looking forward to hearing from you.
Yours Sincerely.

Editor Comments:
1. We note that Figure 1b, 2a 2b and 3a and 3b and Table 5 are not cited in the text of your manuscript. Please ensure that all figures, tables and supplementary/additional files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.
Response: Thank you for your comments, the Figure 1b, 2a 2b and 3a and 3b and Table 5 have cited in the text.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.
Response: We have uploaded the manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.